Author's response to reviews

Title: Outcome Measures of a 6 Minute Walk Test: Relationships with Physiologic and Computed Tomography Findings in Patients with Sarcoidosis

Authors:

Esam H Alhamad (esamalhamad@yahoo.com)
Shaffi Ahmad Shaik (shaffi786@yahoo.com)
Majdy M Idrees (majiidrees@gmail.com)
Mohammed O Alanezi (alanezi@hotmail.com)
Arthur C Isnani (acisnanimd@yahoo.com)

Version: 9 Date: 8 August 2010

Author's response to reviews: see over
Dear Editorial Production team,

RE: Title : Outcome Measures of a 6 Minute Walk Test: Relationships with Physiologic andComputed Tomography Findings in Patients with Sarcoidosis
Journal: BMC Pulmonary Medicine
MS ID : 1801645934294088

Below you will find point-by-point description of the changes made as requested.

We would like to thank the editorial team for their tremendous support and help in facilitating the process of publishing our manuscript

Sincerely,

Esam Alhamad, MD, FCCP, FACP
Assistant Professor of Medicine
Division of Pulmonary Medicine
College of Medicine
King Saud University
Email: esamalhamad@yahoo.com
Major revisions (we require the author to make these changes)

Tables: we notice that you have included the tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references. Alternatively, please cite the files as Additional file 1 etc., and include an additional files section in the manuscript.

Response

Per instruction we have added the tables to the manuscript after references

Minor revisions:

Running title: please remove this from the title page.

Response

Running title was removed

Qualifications: Please remove the authors' qualifications (e.g. PhD / MD) or job titles from the manuscript and submissions system

Response

There are no author's qualifications present in the current manuscript

Keywords: please remove this section.

Response

We removed the keywords

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

Response

All highlighting and tracking have been removed

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance

Response

We checked the manuscript and no typographical errors was found